Author's response to reviews

Title: Word Processing differences between dyslexic and control children

Authors:

Isabella C Paul (Isabella.Paul@uni-konstanz.de)
Christof Bott (Christof.Bott@uni-konstanz.de)
Christian Wienbruch (Christian.Wienbruch@uni-konstanz.de)
Thomas R Elbert (Thomas.Elbert@uni-konstanz.de)

Version: 5 Date: 15 December 2005

Author’s response to reviews: see over
Dear Professor Newmark,

Thank you for accepting our manuscript for publication in BMC Psychiatry. We made the following changes to make the manuscript suitable:

*Title & abstract*

Title & abstract are identical in the manuscript and the submission system.

*Author list*

The list of authors is the same in the manuscript and the submission system.

*Underlining*

All underlining was removed.

*Footnotes*

The footnote was incorporated into the text.

*Figure and table placement text*

The text was removed.

*Authors’ contributions*

Changes were made accordingly.
Thank you again,
Sincerely yours,
Isabella Paul